Author's response to reviews

Title: Enrolling adolescents in HIV vaccine trials: Reflections on legal complexities from South Africa

Authors:

Catherine Slack (slackca@ukzn.ac.za)
Ann Strode (strodea@ukzn.ac.za)
Terry Fleischer (terry@uctgsh1.uct.ac.za)
Glenda Gray (gray@pixie.ac.za)
Chitra Ranchod (ranchod@ukzn.ac.za)

Version: 3 Date: 11 May 2007

Author’s response to reviews:

Affiliations submitted in full as requested and linked to each author by superscript
Authors qualifications removed
Word count and key words removed
Abstract structured according to Background, Discussion and Summary
Manuscripts sections: checked the sections and added "competing interests" in the right order
Authors contributions: Added that all authors read and approved the final manuscript